Author's response to reviews

Title: Effects of Curcuma longa (turmeric) on postprandial plasma glucose and insulin in healthy subjects

Authors:

Jennie Wickenberg (jennie.wickenberg@skane.se)
Sandra Lindstedt Ingemansson (Sandra.Lindstedt@skane.se)
Joanna Hlebowicz (joanna.hlebowicz@med.lu.se)

Version: 4 Date: 8 October 2010

Author's response to reviews:

Dear Editor,

I would like to thank you for your important and valuable comments regarding the format of the present manuscript. We have clearly been able to reply to all your remarks, as follows below. Do not hesitate to contact me further if you have any additional questions. I am hoping for further cooperation in the future.

Please notice that the changes have been done as required.

Major revisions
• Figures: Figure legends 1 and 2 contain now graphics written in text.
• Abstract: the Abstract present in the manuscript matches exactly what is in the submission system. There symbol before the word insulin in the results section of the abstract is now deleted.
• Table 3 is now mentioned in the text.

Minor revisions
• Initials in the email list are now used and not full names.
• The figures are cropped.
• References 12, 13 and 14 in the text are change to square brackets.
• The reference list is corrected.
Please do not hesitate to contact me if you have any questions.

Sincerely,

Joanna Hlebowicz, MD, PhD
Joanna.Hlebowicz@med.lu.se
Phone: +46 40 33 10 00
Fax: +46 40 92 32 72